Author's response to reviews

Title: The PAMINO-project: Evaluating a primary care based educational program to improve the quality of life of palliative patients [ISRCTN78021852]

Authors:

Thomas TR Rosemann (thomas.rosemann@med.uni-heidelberg.de)
Katja KH Hermann (katja.herman@med.uni-heidelberg.de)
Antje AM Miksch (antje.miksch@med.uni-heidelberg.de)
Peter PE Engeser (peter.engeser@med.uni-heidelberg.de)
Joachim JS Szecsenyi (joachim.szecsenyi@med.uni-heidelberg.de)

Version: 4 Date: 24 May 2007

Author's response to reviews:

revisions were performed as required: table 1 was mentioned in the text and the references were revised